Author's response to reviews

Title: A comparative study of the debridement efficacy and apical extrusion of dynamic and passive root canal irrigation systems

Authors:

Ahmed M Alkahtani (ksucod@gmail.com)
Tala D Al Khudhairi (sweetykb@gmail.com)
Sukumaran Anil (drsanil@gmail.com)

Version: 4 Date: 30 January 2014

Author's response to reviews:

January 30, 2014
Ms Eloisa Nolasco
Dr. Giulio Gavini
Journal Editorial Office
BioMed Central

Dear Dr. Gavin
Ref: MS: 4793562611371576
A comparative study of the debridement efficacy and apical extrusion of dynamic and passive root canal irrigation systems; Ahmed M Alkahtani, Tala D Al Khudhairi and Sukumaran Anil

Thank you. The certification document is removed.
Regards
Ahmed Alkahtani